Author's response to reviews

Title: Identification of New Biomarkers for Acute Lung Injury by Expression-based Genome-wide Association Study

Authors:

Dmitry N Grigoryev (dgrigory@umaryland.edu)
Suman Chaudhary (schaudhary@cmh.edu)
Dilyara I Cheranova (dcheranova@cmh.edu)
Daniel P Heruth (dpheruth@cmh.edu)
Li Q Zhang (lqzhang@cmh.edu)
Shui Q Ye (sqye@cmh.edu)

Version: 2 Date: 9 February 2015

Author's response to reviews:

1. Line numbering was added.
2. Email addresses for each author were added to the title page.
3. Acknowledgment section was added (lines 328-329).
4. Ethic consideration for human subjects was added (lines 111-112).